Author's response to reviews

Title: Colorectal cancer screening services in family medicine vs. internal medicine practices: a cross-sectional study

Authors:

Angela Y. Higgins (angela.higgins@umassmed.edu)
Anna R. B. Doubeni MD, MPH (Anna.Doubeni@umassmed.edu)
Karon L. Phillips PhD, MPH (karonlphillips@hotmail.com)
Adeyinka O. Laiyemo MD, MPH (adeyinka.laiyemo@Howard.edu)
Becky Briesacher PhD (Becky.Briesacher@umassmed.edu)
Jennifer Tjia MD, MSc (Jennifer.Tjia@umassmed.edu)
Chyke A. Doubeni MD, MPH (Chyke.Doubeni@umassmed.edu)

Version: 2 Date: 5 December 2011

Author's response to reviews:

Chyke A. Doubeni, MD, MPH
Assistant Professor of Family Medicine and Community Health
University of Massachusetts Medical School
55 N. Lake Avenue
Worcester, MA 01655
Phone (o): 774-443-7772; (c): 508-579-3934
Fax: 774-441-6212
Email: chyke.doubeni@umassmed.edu

December 5, 2011

Mark Andrew Cardinez
BMC Gastroenterology
BioMed Central Ltd.
236 Gray's Inn Road
London, WC1X 8HB, United Kingdom

Dear Mr. Cardinez:

Thank you for giving us the opportunity to make this correction. I have made the required correction to the manuscript titled: “Colorectal cancer screening services...
in family medicine vs. internal medicine practices: a cross-sectional study” submitted for publication in BMC Gastroenterology.

I have added the phrase, “this study was reviewed by the institutional review board (IRB) of the University of Massachusetts Medical School (Worcester, MA) and was considered exempt from a full IRB review” to the end of the first paragraph of the methods section.

Thank you again for considering our manuscript for publication and we look forward to hearing from you.

Sincerely,

Chyke A. Doubeni, MD, MPH